Author’s response to reviews

Title: Gut microbiota in children with type 1 diabetes differs from healthy children: a case-control study.

Authors:

Mora M Murri (moramurri@gmail.com)
Isabel I Leiva (isabeleiva@hotmail.com)
Juan Miguel JM Gomez-Zumaquero (juanm.gomez.exts@juntadeandalucia.es)
Fernando F Cardona (fernandocardonadiaz@gmail.com)
Federico F Soriguer (federico.soriguer.sspa@juntadeandalucia.es)
Francisco J FJ Tinahones (fjtinahones@hotmail.com)
María Isabel MI Queipo-Ortuño (maribelqo@gmail.com)

Version: 4 Date: 14 January 2013

Author’s response to reviews:

Laboratorio Investigación Biomédica 1ª planta.
Hospital Clínico Universitario Virgen de la Victoria.
Campus de Teatinos s/n 29010 Málaga, Spain.
Tel: +34 951032647 / 48
Fax: +34 951924651
e-mail : maribelqo@gmail.com

The Editor,
BMC Medicine
14 January 2012
Ref: Gut microbiota in children with type 1 diabetes differs from healthy children: a case-control study (MS ID: 1845900297788147)

Dear Editor
Thank you very much for accepting our manuscript for publication in the journal. We set out below our responses to the comments of the Editorial Production Team, together with the action taken.

Yours sincerely,
Dr. María Isabel Queipo-Ortuño

Major revisions (we require the author to make these changes)
Comment 1: Please ensure that each author affiliation contains, in the following order, department, institution, street name and number, town, post code and country.
Response: The affiliations of all authors of the manuscript have been made as it has been indicated.

Comment 2: Please cite references using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop, e.g. "text [1]."
Response: The references have been cited using consecutive numbers in brackets and placed before the full stop through the manuscript.

Comment 3: Please remove all issue numbers from the reference list.
Response: All issue numbers have been removed from the reference list.

Comment 4: Please provide a brief title of no more than 15 words for figure 1. This should be first sentence of the figure legends. The legend for the parts of the figure should follow from this.
Response: A short title has been added to the figure 1.

Comment 5: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: We have checked our manuscript to eliminate any typographical errors.